Author's response to reviews

Title: What Motivates Senior Clinicians To Teach Medical Students?

Authors:

Jane Dahlstrom (jane.dahlstrom@act.gov.au)
Anna Dorai-Raj (dr_anna_dorai_raj@hotmail.com)
Darryl McGill (darryl.mcgill@act.gov.au)
Cathy Owen (cathy.owen@anu.edu.au)
Kathleen Tymms (canberra_rheumatology@bigpond.com)
D. Ashley R Watson (Ashley.Watson@acl.gov.au)

Version: 3 Date: 11 May 2005

Author's response to reviews:

Dear Reviewers
thanks for your comments they have improved the quality of the paper- all suggestions have been adopted (all compulsory ones were). The only exception was the decision to stick with the use of the term "ipsative data". The access date was always in the reference when toggled. Final version is uploaded below

Thanks The Motivators